w DO Status & Plans

® Experiment is in good shape, and basically ready for beam
¢ Shutdown status has been presented several times already
® Detector fully closed last Wednesday
® Wednesday evening magnet tests
¢ Toroid powered up fine with both polarities
¢ Initial problem with solenoid quickly solved by replacing current module
® Thursday and Friday: preparation and recovery from 2 power outages
¢ No problems recovering from power outages

¢ Delaying Thursday power outage from ~12:00 to 17:00 was major annoyance, lost
time for literally hundreds of people

® Established zero-bias running with complete detector on Saturday and
Sunday, cosmics with entire detector Sunday evening

® Completed supervised access on Sunday evening

® Expect — as usual — brief controlled accesses for minor electronics repairs

¢ In particular, some CFT/PS AFE boards are still undergoing debugging

Arnd Meyer (RWTH Aachen) Nov 17,2003 Page 1



w DO Status & Plans

® Extended cosmic ray running and special runs for the next couple of days

¢ DAQ is not yet as stable as it should be - lots of underlying soft-/hardware
changed, shake down DAQ and trigger

¢ Find as many detector problems as possible before first store
¢ Continue 24x7 shifts, detector shifters on-call during owl shifts
® Will start up with pre-shutdown trigger list to establish baseline

® After successful tests, change prescales to make use of expected higher
readout bandwidth (1 kHz) into Level 3

From the beginning running with 5 of 6 Silicon Track Trigger crates in the
readout

All Beta Level 2 system
Factor of 2 to 10 reduction in calorimeter noise
Full luminosity system readout

Completed FPD system (detectors and readout)

Some commissioning effort expected to integrate STT, FPD, L2, luminosity
system, preshower
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